Author's response to reviews

Title: Valsalva retinopathy in pregnancy: a case report

Authors:

Abdullah S Al-Mujaini (mujainisqu@hotmail.com)
Carolina C Montana (caro_montana@yahoo.com)

Version: 5 Date: 1 April 2008

Author's response to reviews:

Apr. 01. 2008
To:
Dr. Michael Kidd
Editor- In- Chief
Journal of Medical Case Report

Dear Dr. Kidd;
Sub: Submission of revised version of manuscript “Valsalva Retinopathy In Pregnancy: a case report ”

We are submitting the revised version of the above-mentioned manuscript as per the editorial production team request. We have gone through the following changes;
1-The author’s names were updated in the manuscript and written exactly as they are in the submission system.
2-The title “Background“ has been changed to “Introduction” in the abstract and manuscript.
3-The consent section has been placed after the author’s contribution
4-All these alterations have been made in www.jmedicalcasereports.com/media/man/2069580620167459/msd/main/0

With regards,
Yours sincerely,

Abdullah Al-Mujaini, MD, FRCSC
Corresponding author